% Generated by IEEEtran.bst, version: 1.13 (2008/09/30)
\begin{thebibliography}{10}
\providecommand{\url}[1]{#1}
\csname url@samestyle\endcsname
\providecommand{\newblock}{\relax}
\providecommand{\bibinfo}[2]{#2}
\providecommand{\BIBentrySTDinterwordspacing}{\spaceskip=0pt\relax}
\providecommand{\BIBentryALTinterwordstretchfactor}{4}
\providecommand{\BIBentryALTinterwordspacing}{\spaceskip=\fontdimen2\font plus
\BIBentryALTinterwordstretchfactor\fontdimen3\font minus
  \fontdimen4\font\relax}
\providecommand{\BIBforeignlanguage}[2]{{%
\expandafter\ifx\csname l@#1\endcsname\relax
\typeout{** WARNING: IEEEtran.bst: No hyphenation pattern has been}%
\typeout{** loaded for the language `#1'. Using the pattern for}%
\typeout{** the default language instead.}%
\else
\language=\csname l@#1\endcsname
\fi
#2}}
\providecommand{\BIBdecl}{\relax}
\BIBdecl

\bibitem{Frank+Asuncion:2010}
\BIBentryALTinterwordspacing
A.~Frank and A.~Asuncion, ``{UCI} machine learning repository,'' 2010.
  [Online]. Available: \url{http://archive.ics.uci.edu/ml}
\BIBentrySTDinterwordspacing

\bibitem{Hart67}
T.~Cover and P.~Hart, ``Nearest neighbor pattern classification,''
  \emph{Information Theory, IEEE Transactions on}, vol.~13, no.~1, pp. 21--27,
  Jan 1967.

\bibitem{jalali08}
O.~Jalali, T.~Menzies, and M.~Feather, ``Optimizing requirements decisions with
  keys,'' in \emph{Proceedings of the PROMISE 2008 Workshop (ICSE)}, 2008,
  available from \url{http://menzies.us/pdf/08keys.pdf}.

\bibitem{Devi2002}
\BIBentryALTinterwordspacing
V.~S. Devi and M.~N. Murty, ``An incremental prototype set building
  technique,'' \emph{Pattern Recognition}, vol.~35, no.~2, pp. 505 -- 513,
  2002. [Online]. Available:
  \url{http://www.sciencedirect.com/science/article/B6V14-44HT45G-K/2/a27d8d6d%
4a216b97974cfa5cd7947419}
\BIBentrySTDinterwordspacing

\bibitem{Bezdek2001}
J.~Bezdek and L.~Kuncheva, ``{Nearest prototype classifier designs: An
  experimental study},'' \emph{{INTERNATIONAL JOURNAL OF INTELLIGENT SYSTEMS}},
  vol.~{16}, no.~{12}, pp. {1445--1473}, {DEC} {2001}.

\bibitem{Dasarathy94}
B.~Dasarathy, ``Minimal consistent set (mcs) identification for optimal nearest
  neighbor decision systems design,'' \emph{Systems, Man and Cybernetics, IEEE
  Transactions on}, vol.~24, no.~3, pp. 511--517, Mar 1994.

\bibitem{Li2009}
Y.~Li, M.~Xie, and T.~Goh, ``{A study of project selection and feature
  weighting for analogy based software cost estimation},'' \emph{Journal of
  Systems and Software}, vol.~82, pp. 241--252, 2009.

\bibitem{Bezdek98}
J.~Bezdek, T.~Reichherzer, G.~Lim, and Y.~Attikiouzel, ``Multiple-prototype
  classifier design,'' \emph{Systems, Man, and Cybernetics, Part C:
  Applications and Reviews, IEEE Transactions on}, vol.~28, no.~1, pp. 67--79,
  Feb 1998.

\bibitem{Cano2005}
\BIBentryALTinterwordspacing
J.~R. Cano, F.~Herrera, and M.~Lozano, ``Stratification for scaling up
  evolutionary prototype selection,'' \emph{Pattern Recognition Letters},
  vol.~26, no.~7, pp. 953 -- 963, 2005. [Online]. Available:
  \url{http://www.sciencedirect.com/science/article/B6V15-4DTTHDM-1/2/1c98d61e%
651bcd792d23fce1360b91c9}
\BIBentrySTDinterwordspacing

\bibitem{Garain2008}
U.~Garain, ``Prototype reduction using an artificial immune model,''
  \emph{Pattern Anal. Appl.}, vol.~11, no. 3-4, pp. 353--363, 2008.

\bibitem{Veenman2005}
C.~Veenman and M.~Reinders, ``{The nearest subclass classifier: A compromise
  between the nearest mean and nearest neighbor classifier},'' \emph{{IEEE
  TRANSACTIONS ON PATTERN ANALYSIS AND MACHINE INTELLIGENCE}}, vol.~{27},
  no.~{9}, pp. {1417--1429}, {SEP} {2005}.

\bibitem{García2008}
\BIBentryALTinterwordspacing
S.~García, J.~R. Cano, and F.~Herrera, ``A memetic algorithm for evolutionary
  prototype selection: A scaling up approach,'' \emph{Pattern Recognition},
  vol.~41, no.~8, pp. 2693 -- 2709, 2008. [Online]. Available:
  \url{http://www.sciencedirect.com/science/article/B6V14-4RX06Y7-1/2/af922a53%
032b4fa079170bd319749e2e}
\BIBentrySTDinterwordspacing

\bibitem{Hart68a}
P.~Hart, ``The condensed nearest neighbor rule (corresp.),'' \emph{Information
  Theory, IEEE Transactions on}, vol.~14, no.~3, pp. 515 -- 516, may 1968.

\bibitem{Gates72}
G.~Gates, ``The reduced nearest neighbor rule (corresp.),'' \emph{Information
  Theory, IEEE Transactions on}, vol.~18, no.~3, pp. 431 -- 433, 1972.

\bibitem{lot2010}
\BIBentryALTinterwordspacing
J.~Olvera-López, J.~Carrasco-Ochoa, and J.~Martínez-Trinidad, ``A new fast
  prototype selection method based on clustering,'' \emph{Pattern Analysis
  &amp; Applications}, vol.~13, pp. 131--141, 2010, 10.1007/s10044-008-0142-x.
  [Online]. Available: \url{http://dx.doi.org/10.1007/s10044-008-0142-x}
\BIBentrySTDinterwordspacing

\bibitem{Olvera2010}
J.~Olvera-López, J.~Carrasco-Ochoa, J.~Martínez-Trinidad, and J.~Kittler, ``A
  review of instance selection methods,'' \emph{Artificial Intelligence
  Review}, vol.~34, pp. 133--143, 2010.

\bibitem{FCNN07}
F.~Angiulli, ``{Fast nearest neighbor condensation for large data sets
  classification},'' \emph{{IEEE TRANSACTIONS ON KNOWLEDGE AND DATA
  ENGINEERING}}, vol.~{19}, no.~{11}, pp. {1450--1464}, {NOV} {2007}.

\bibitem{wilson00}
D.~R. Wilson and T.~R. Martinez, ``Reduction techniques for instance-based
  learning algorithms,'' \emph{Machine Learning}, vol.~38, pp. 257--286, 2000.

\bibitem{Ritter75}
G.~Ritter, H.~Woodruff, S.~Lowry, and T.~Isenhour, ``An algorithm for a
  selective nearest neighbor decision rule (corresp.),'' \emph{Information
  Theory, IEEE Transactions on}, vol.~21, no.~6, pp. 665 -- 669, nov 1975.

\bibitem{Wilson72}
D.~L. Wilson, ``Asymptotic properties of nearest neighbor rules using edited
  data,'' \emph{Systems, Man and Cybernetics, IEEE Transactions on}, vol.~2,
  no.~3, pp. 408 -- 421, jul 1972.

\bibitem{Raicharoen05}
T.~Raicharoen and C.~Lursinsap, ``{A divide-and-conquer approach to the
  pairwise opposite class-nearest neighbor (POC-NN) algorithm},''
  \emph{{PATTERN RECOGNITION LETTERS}}, vol.~{26}, no.~{10}, pp. {1554--1567},
  {JUL 15} {2005}.

\bibitem{Tomek76}
I.~TOMEK, ``{EXPERIMENT WITH EDITED NEAREST-NEIGHBOR RULE},'' \emph{{IEEE
  TRANSACTIONS ON SYSTEMS MAN AND CYBERNETICS}}, vol.~{6}, no.~{6}, pp.
  {448--452}, {1976}.

\bibitem{Kriegel05}
H.~peter Kriegel, A.~Pryakhin, and M.~Schubert, ``Multi-represented
  knn-classification for large class sets,'' in \emph{10 th Intl. Proc}, 2005.

\bibitem{Brighton2002}
H.~Brighton and C.~Mellish, ``{Advances in instance selection for
  instance-based learning algorithms},'' \emph{{DATA MINING AND KNOWLEDGE
  DISCOVERY}}, vol.~{6}, no.~{2}, pp. {153--172}, {APR} {2002}.

\bibitem{OCM2007}
J.~Arturo Olvera-López, J.~Ariel Carrasco-Ochoa, and J.~Francisco
  Martínez-Trinidad, ``Object selection based on clustering and border
  objects,'' in \emph{Computer Recognition Systems 2}, ser. Advances in
  Intelligent and Soft Computing, M.~Kurzynski, E.~Puchala, M.~Wozniak, and
  A.~Zolnierek, Eds.\hskip 1em plus 0.5em minus 0.4em\relax Springer Berlin /
  Heidelberg, 2007, vol.~45, pp. 27--34.

\bibitem{Carrasco05}
J.~Olvera-Lopez, J.~Carrasco-Ochoa, and J.~Martinez-Trinidad, ``{Sequential
  search for decremental...}'' in \emph{{INTELLIGENT DATA ENGINEERING AND
  AUTOMATED LEARNING IDEAL 2005, PROCEEDINGS}}, ser. {LECTURE NOTES IN COMPUTER
  SCIENCE}, {Gallagher, M and Hogan, J and Maire, F}, Ed., {2005}, vol. {3578},
  pp. {280--285}.

\bibitem{Riquelme03}
J.~Riquelme, J.~Aguilar-Ruiz, and M.~Toro, ``{Finding representative patterns
  with ordered projections},'' \emph{{PATTERN RECOGNITION}}, vol.~{36},
  no.~{4}, pp. {1009--1018}, {APR} {2003}.

\bibitem{Narayan06}
B.~Narayan, C.~Murthy, and S.~Pal, ``{Maxdiff kd-trees for data
  condensation},'' \emph{{PATTERN RECOGNITION LETTERS}}, vol.~{27}, no.~{3},
  pp. {187--200}, {FEB} {2006}.

\bibitem{knn}
R.~O. Duda and P.~E.Hart, \emph{Pattern classification and scene
  analysis}.\hskip 1em plus 0.5em minus 0.4em\relax A Wiley-Interscience
  Publication, New York: Wiley, 1973.

\bibitem{Fayyad1992}
U.~M. Fayyad and K.~B. Irani, ``On the handling of continuous-valued attributes
  in decision tree generation,'' \emph{Mach. Learn.}, vol.~8, pp. 87--102,
  January 1992.

\bibitem{burak}
B.~Turhan, T.~Menzies, A.~B. Bener, and J.~Di~Stefano, ``{On the relative value
  of cross-company and within-company data for defect prediction},''
  \emph{{EMPIRICAL SOFTWARE ENGINEERING}}, vol.~{14}, no.~{5}, pp. {540--578},
  {OCT} {2009}.

\bibitem{Walsh94}
K.~Walsh, J.~Buckleton, and C.~Triggs, ``A practical example of the
  interpretation of glass evidence,'' \emph{Science and Justice}, vol.~36,
  no.~4, pp. 213--218, 1996.

\bibitem{09NAS}
C.~on~Identifying the Needs of the Forensic Sciences Community;Committee~on
  Applied and N.~R.~C. Theoretical~Statistics, \emph{Strengthening Forensic
  Science in the United States: A Path Forward}.\hskip 1em plus 0.5em minus
  0.4em\relax 500 Fifth Street, N.W., Lockbox 285, Washington, DC 20055:
  National Academies Press, 2009.

\bibitem{Karslake09}
E.~Liszewski, S.~Lewis, J.~Siegel, and J.~Goodpaster, ``Characterisation of
  automotive paint clear coats by ultraviolet absorption microspectrophotometry
  with subsequent chemometric analysis,'' \emph{Applied Spectroscopy}, vol.~64,
  no.~10, pp. 1122--1125, 2010.

\bibitem{fastmap}
C.~Faloutsos and K.-I. Lin, ``Fastmap: a fast algorithm for indexing,
  data-mining and visualization of traditional and multimedia datasets,'' in
  \emph{SIGMOD '95: Proceedings of the 1995 ACM SIGMOD international conference
  on Management of data}.\hskip 1em plus 0.5em minus 0.4em\relax New York, NY,
  USA: ACM, 1995, pp. 163--174.

\bibitem{09Zadora}
G.~Zadora and T.~Neocleous, ``{Likelihood ratio model for classification of
  forensic evidence},'' \emph{{ANALYTICA CHIMICA ACTA}}, vol. {642}, no. {1-2,
  Sp. Iss. SI}, pp. {266--278}, {MAY 29} {2009}.

\bibitem{09aZadora}
G.~Zadora, ``{Evaluation of evidence value of glass fragments by likelihood
  ratio and Bayesian Network approaches},'' \emph{{ANALYTICA CHIMICA ACTA}},
  vol. {642}, no. {1-2, Sp. Iss. SI}, pp. {279--290}, {MAY 29} {2009}.

\bibitem{06Aitken}
C.~Aitken, D.~Lucy, G.~Zadora, and J.~Curran, ``{Evaluation of transfer
  evidence for three-level multivariate data with the use of graphical
  models},'' \emph{{COMPUTATIONAL STATISTICS \& DATA ANALYSIS}}, vol.~{50},
  no.~{10}, pp. {2571--2588}, {JUN 20} {2006}.

\bibitem{04Aitken}
C.~Aitken and D.~Lucy, ``{Evaluation of trace evidence in the form of
  multivariate data},'' \emph{{JOURNAL OF THE ROYAL STATISTICAL SOCIETY SERIES
  C-APPLIED STATISTICS}}, vol.~{53}, no. {Part 1}, pp. {109--122}, {2004}.

\bibitem{02Koons}
R.~Koons and J.~Buscaglia, ``{Interpretation of glass composition measurements:
  the effects of match criteria on discrimination capability},'' \emph{{JOURNAL
  OF FORENSIC SCIENCES}}, vol.~{47}, no.~{3}, pp. {505--512}, {MAY} {2002}.

\bibitem{99Koons}
------, ``{The forensic significance of glass composition and refractive index
  measurements},'' \emph{{JOURNAL OF FORENSIC SCIENCES}}, vol.~{44}, no.~{3},
  pp. {496--503}, {MAY} {1999}.

\bibitem{me07}
T.~Menzies and Y.~Hu, ``Just enough learning (of association rules): The {TAR2}
  treatment learner,'' in \emph{Artificial Intelligence Review}, 2007,
  available from \url{http://menzies.us/pdf/07tar2.pdf}.

\end{thebibliography}

%notes - instead of reduction % use storage requirements
% find the most popular of the IS and use those in the table


\section{CLIFF Assessment}
\label{section:assess}
In Section \ref{section:cliff}, we discussed the design and operation of CLIFF with the help of a simple example. We also showed its time complexity as linear - $O(n)$ in Section \ref{section:time}. In this section, we look at how CLIFF performs against other IS with three(3) evaluation methods. CLIFF is evaluated by first comparing its performance with three(3) IS spanning the decades from 1968 (Hart's CNN \cite{Hart68a}) to MCS in 1994 \cite{Dasarathy94} and finally 2010 PSC \cite{lot2010}. We then move on to examine the noise tolerance of each IS studied here by introducing artificial noise to the training sets. Finally we take a look at what we call the $brittleness$ measure. Brittleness, is defined as \emph{a tiny change in the input data can lead to a major change in the output}. As mentioned previously, we view instance selection as a viable method to decrease $brittleness$ and the following sections will show that CLIFF does a better job of reducing the impact of $brittleness$ than other IS evaluated in this work.

%\section{CLIFF Assessment on Standard Data Sets}
%\label{section:assess}
\subsection{Data Sets}
\fig{info} lists the nine(9) data sets used to assess CLIFF. The number of instances and attributes per instance are shown for each data set, along with the number of distinct classes of instances. All of these data sets were acquired from the UCI repository \cite{Frank+Asuncion:2010}. Since CLIFF is designed for data sets with discrete values, all data sets with numeric attribute values are discretized by the Fayyad and Irani \cite{Fayyad1992} discretizer implemented in WEKA. 

\begin{figure}
\begin{center}
\scalebox{0.8}{
\begin{tabular}{ | l | l | l | l |}
\hline
Data & Instances & Attributes & Class \\ \hline
Breast Cancer (bcancer) & 286 & 9 & 2 \\ \hline
Ecoli (ec) & 366 & 8 & 8 \\ \hline
Glass (gl) & 214 & 9 & 6 \\ \hline
Heart (ht) & 303 & 13 & 2 \\ \hline
Hepatitis (hep) & 155 & 19 & 2 \\ \hline
Iris (ir) & 150 & 4 & 3 \\ \hline
Labor (lb) & 57 & 16 & 2 \\ \hline
Lymph (ly)& 148 & 18 & 4 \\ \hline
Vote (vt) & 435 & 16 & 2 \\ \hline
\end{tabular}}
\end{center}
\caption{Data Set Characteristics}
\label{fig:info}
\end{figure}     

\subsection{Experimental Method}
\label{section:brit}

We evaluate CLIFF as an instance selector on standard data sets in cross validation experiments. Its performance compared with KNN, CNN, MCS and PSC is characterized using five(5) measures of evaluation:
\be
\item using probability of detection (pd);
\item precision (prec); 
\item probability of false alarm (pf);
\item brittleness (brit);
\item reduction percent (red) which can indicate noise tolerance.
\ee

One(1) to three(3) above is completed as follows \cite{burak}: By allowing A, B, C and D to represent true negatives, false negatives, false positives and true positives respectfully, it then follows that \emph{pd} also known as recall, is the result of true positives divided by the sum of false negative and true positives \emph{D / (B + D)}. While pf is the result of: \emph{C / (A + C)}. Finally, precision is the result of: \emph{D / (C + D)}. The $pd$, $prec$ and $pf$ values range from 0 to 1. When there are no false alarms $pf$ = 0 and at 100\% detection, $pd$ = 1.

Next, brittleness is measured by distance each test instance moves before changing its original predicted target class. Intuitively, the further away the test instance has to move the less $brittle$ the model. Collecting the distances of each test instance with a predicted target class to the nearest training instance with a different target class. This is done for all IS and knn where $k=1$. The distances generated are joined, sorted and labeled according to their position in the list. For example, let us say that A and B are IS with the distance values of \{2, 2, 2, 3, 4, 3, 77\} and \{6, 7, 3, 9, 1, 1, 1, 100\} respectively. After being joined, sorted and labeled the result is as follows:

\begin{figure}[ht!]
\scalebox{0.7}{
\begin{tabular}{l*{14}{c}r}
IS  & B & B & B & A  & A & A & A & A & B &  A &  B &  B &  B &  A  &  B \\
\hline
Sort & 1 & 1 & 1 & 2 & 2 & 2 & 3 & 3 & 3 &  4 &  6 &  7 &  9 & 77 & 100  \\
Position & 1 & 2 & 3 & 4 & 5 & 6 & 7 & 8 & 9 & 10 & 11 & 12 & 13 & 14 &  15  \\
Label & 1 & 1 & 1 & 5 & 5 & 5 & 8 & 8 & 8 & 10 & 11 & 12 & 13 & 14  & 15  \\
\end{tabular}}
\end{figure}
	
As shown, labels are assigned according to the position of a value in the list, however, if values are the same, the mean of their position values are used as a label.

Last, the reduction percent is simply the number of the instances selected as compared to the original training data. This measure is particularly important in terms of noise. According to \cite{wilson00}, in the presence of class noise very few instances will be removed from the training set thus increasing the reduction percent. Using the class noise injector described in \cite{wilson00}, noise is introduced to training data by randomly changing the target class of \{10, 20, 30, 40\}\% of the instances in the training data to any other target class value.

Our experiment design follows the pseudo code given in \fig{knnexp1} for the standard data sets in \fig{info}. For each data-set, tests were built from 20\% of the data, selected at random. Instances were then selected by an instance selector from the remaining 80\% of the data. Next, models were learned from the selected instances using the {\em 1 nearest neighbor rule} and the euclidean distance as the distance metric.

This procedure was repeated 5 times, randomizing the order of data in each data-set each time. In the end CLIFF is tested and trained 25 times for each data-set. 

\begin{figure}[ht!]
\renewcommand{\baselinestretch}{1}
\begin{algorithmic} \footnotesize
\STATE {\em data = [breastcaner ecoli glass heart hepatitis iris labor lymph vote]}
\STATE {\em learner = [knn]}
\STATE {\em IS = [knn cliff cnn mcs psc]}
\STATE {\em noise = [10 20 30 40]}
\STATE {\em brit = list of distances: \{D(predicted test) - D(NUN)\}}
\STATE {\em red = reduction percentage}
%\STATE {\em $stat_{test}$ = [Mann Whitney]}
\STATE {\em count = 0}

\FORALL {~IS}
  \REPEAT
    \FORALL {~d in data}
     \FORALL {~n in noise}
     \STATE train = random 80\% of data
     \STATE test = d - train
		
     \COMMENT {Construct model from train data}
     \STATE $r_{train}$ = Reduce train with IS
     \STATE model = Train learner with $r_{train}$
     
     \COMMENT {Evaluate model on test data}
     \STATE [pd, prec, pf, red, brit] = model on test
   \ENDFOR
   \ENDFOR
   \STATE {\em count = count + 1}
  \UNTIL {\em count = 5}
\ENDFOR  	
\end{algorithmic}
\caption{Pseudo code for Experiment}
\label{fig:knnexp1}
\end{figure}

\begin{figure*}
  \centering
  \subfloat{\label{}\includegraphics[width=0.195\textwidth]{red0}}                
  \subfloat{\label{}\includegraphics[width=0.195\textwidth]{red1}}
  \subfloat{\label{}\includegraphics[width=0.195\textwidth]{red2}}
  \subfloat{\label{}\includegraphics[width=0.195\textwidth]{red3}}
  \subfloat{\label{}\includegraphics[width=0.195\textwidth]{red4}}
  \caption{Effect of noise on reduction percent. As noise increases from 0 - 40\%, inferior instance selectors generally increase in reduction \%, i.e. the number of instances selected from each training set increases as noise increases.}
  \label{fig:red-results}
\end{figure*}

\begin{figure}[ht!]
\begin{center}
\includegraphics[scale=1]{breastcancer}
\end{center}
\caption{Visual illustration of brittleness reduction in the breast cancer data-set for KNN and the IS. Here, the greater the position of each test instance with a predicted target class from an instance with a different target class the better.}\label{fig:brit-results}
\end{figure}


\begin{comment}
\begin{figure*}
  \centering
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{breastcancer}}                
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{ecl}}
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{glass}} \\
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{hc}}
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{hepatitis}}
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{iris}} \\
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{labor}}
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{lymph}}
  \subfloat{\label{}\includegraphics[width=0.3\textwidth]{vote}}

  \caption{Visual illustration of brittleness}
  \label{fig:brit-results}
\end{figure*}




%The results were visualized using \emph{quartile charts} as in \cite{burak}. To generate these charts the performance measures for the $pd$s and $pf$s are sorted to get the median, lower and upper quartile of numbers. For our quartile charts, the upper and lower quartiles are marked with black lines; the median is marked with a black dot; and the vertical bars are added to mark the 50\% percentile value. \fig{qc} shows an example where the upper and lower quartiles are 39\% and 59\% respectively, while the median is 49\%. 

\begin{figure}[ht!]
  \begin{center}
  \scalebox{0.5}{
    \begin{tabular}{l}
      \resizebox{100mm}{!}{\includegraphics{quartc}}      
    \end{tabular}}
    \caption{}
    \label{fig:qc}
  \end{center}
\end{figure}
\end{comment}

%Finally, the Mann-Whitney U test was used to test for statistical difference of the different IS. These results are shown as rank values starting at one(1). The lower the rank value, the better the performance of the instance selector. Please note that IS with the same rank value, are not statistically different. The following sections describes the experiments and discusses the results.

%\subsection{Is CLIFF viable as an IS for NN?}

\subsection{Experimental Results}

To answer the question: Is CLIFF viable as an instance selector for NN? We first look at the results for reduction percent. Second, we provide a visual illustration regarding the reduction of brittleness by each instance selector studied. Third, look at the effects of noise on the performance (pd, prec and pf) of KNN and CLIFF. Finally, we determine if CLIFF is statistically better than KNN and the 3 IS studied here.


\subsubsection{Reduction Percent}
This measure addresses the issue of large storage requirement for the NN algorithm. Additionally, it also gives an indication of the noise tolerance of different IS and NN.

\fig{red-results}, visually illustrates the reduction percent results for each data-set. Each chart reports at different noise levels from zero to 40\%. When there is no noise present, all the IS reduces the training set to at most 60\%. CLIFF is impressive, with its reduction percent at 5\% and below for all data-sets except for Ecoli where it got as high as 47\%. As noise increases CLIFF reports generally stable results while the other instance selectors show a gradual increase in their reduction size. Therefore, in addition to a small storage requirement, these results clearly indicate that unlike the other IS, CLIFF is resistant to noise.


\subsubsection{Visualization of Brittleness}
Recall our contention that IS are a feasible way to reduce brittleness in NN. In \fig{brit-results} the pattern is very clear. CLIFF does a much better job of reducing $brittleness$ in all cases than any of the other IS. \fig{brit-results} represents the result for the breast-cancer data-set used in this work. This result is generally replicated in the other eight data-sets used in this work. They all show that the CLIFF test instances have to move further away (most of the time) before there is a change in their target classes. These results are confirmed by a Mann Whitney statistical test, which reveal that the CLIFF brittleness results are statistically different and better than those of the other IS. 

\subsubsection{The Effects of Noise}

The goal here is to see if the CLIFF works well in the presence of noise. This is important because according to \cite{wilson00}, the performance of an instance selector can suffer especially if noisy instances are kept while good instances are removed. In \fig{noise} we compare the performance (pd, prec and pf) of CLIFF to KNN. In the presence of noise we expect CLIFF performance measures to remain study while those of KNN degrade significantly. The results in \fig{noise} corroborates this intuition. For KNN as noise increases both the $pd$ and $prec$ measures decrease. This is especially true for the Ecoli, Glass and Iris data-sets whose measures show a drastic reduction when 20\% class noise is introduced. On the other hand, the CLIFF measures tend to remain steady as noise increases. The exceptions to this are the Hepatitis, Labor and Lymph data-sets. 

\subsubsection{Is CLIFF significantly better than other IS?}

\fig{better1} to \fig{better3} summarizes the results of this experiment. Here, the $pd$, $prec$ and $pf$ measures of KNN, MCS and PSC are subtracted from the corresponding CLIFF measures showing whether or not their results are better the CLIFF. Note that missing values indicate that the result was not significantly different to CLIFF according to the Mann-Whitney U test. The $pd$ measures in \fig{better1} and the $prec$ measures in \fig{better2}, show CLIFF outperforming all other IS and KNN for all data-sets except for Labor when no noise is included. Similarly, in \fig{better3}, CLIFF shows its superiority of all other IS for $pf$ measures, except in the case of Labor at 0\% noise.



%The goal here is to see if the performance of CLIFF is comparable or better than the plain k nearest neighbor (KNN) algorithm, and the CNN, MCS and PSC instance selectors. 

%In this experiment we compare the performance of predicting the target class using the entire training set to using only the instances selected by the IS including CLIFF. To accomplish this, our experiment design follows the pseudo code given in \fig{knnexp1} for the standard data sets. For each data set, tests were built from 20\% of the data, selected at random. Instances are then selected by an instance selector from the remaining 80\% of the data. Next, models were learned from the selected instances.

%This procedure was repeated 5 times, randomizing the order of data in each data-set each time. In the end CLIFF is tested and trained 25 times for each data set. 



%\subsubsection{Results from Experiment \#1}
%The results for this experiment and Experiment 2 (discussed in the following section) are shown in \fig{res1}. For each data-set, the first column shows results where no noise is added to the training set. the each figure shows the results for a $clean$ data-set (without noise) and a $noisy$ data-set (with noise). For both the clean and noisy data-sets, the [pd, pf] quartile charts, the percentage of the data-set used for training after using the IS and the rank values showing the significant difference between the IS and KNN are presented. 

%Let us focus on the $clean$ results for each data-set from \fig{info}. First, we compare the CLIFF results with those of the baseline KNN. As shown, despite using only 9\% to 15\% of the training set, the pds and pfs results of CLIFF compares favorably with those of KNN showing similar or better rank values in most cases. For example, the Mammography(mm) data set (\fig{results106}) ranks as number one(1) for both the pd and pf results while KNN ranks at number three(3). In \fig{results105}, the Liver(lv) data set exhibits statistically similar results for pd while for pf CLIFF has a much better statistical performance than KNN. \fig{results101} to \fig{results104} also show encouraging results for CLIFF as compared with KNN, however \fig{results100}, the Breast Cancer(bc) data set shows an exception, with the pd and pf statistical results for KNN better than CLIFF.

%Next, let us consider the results of the IS. As compared with the other IS, CLIFF does as well as or better than the others in all but one(1) case. However, although for the $bc$ data set, CNN present statistically better pd and pf results, it does so with 62\% of the training data for median pd and pf values of 55\% and 39\% respectively, while CLIFF only needed 11\% of the training data for median pd and pf values of 67\% and 20\% respectively.   

%So generally, CLIFF has markedly lower pfs than the other IS and KNN, this is especially true for the $hh$ and $ir$ data sets. Also the low pfs does not come at the cost of lower pds, infact, CLIFF's median pd results are competitive (sometimes even the best) of all the other IS. 

\begin{comment}
\begin{figure}
\begin{center}
\scalebox{0.9}{
\begin{tabular}{p{2cm} | l | r | r | r}
\hline
UCI Data  & IS & pd & prec & pf \\
\hline
\multirow{5}{BreastCancer} & knn & 62.5 & 66.7 & 35.1 \\
  & knn+cliff & 58.3 & 66.0 & 35.5 \\
  & knn+cnn & 63.2 & 63.6 & 35.0 \\
  & knn+mcs & 52.9 & 52.4 & 46.7 \\ 
  & knn+psc & 50.0 & 48.0 & 48.8 \\ \hline
\multirow{5}{Ecoli} & knn & 62.5 & 66.7 & 1.6  \\
  & knn+cliff & 57.1 & 75.0 & 1.7 \\
  & knn+cnn & 33.3 & 37.5 & 1.7 \\
  & knn+mcs & 33.3 & 31.2 & 1.6 \\ 
  & knn+psc & 20.0 & 26.9 & 3.0  \\ \hline
\multirow{5}{Glass} & knn & 66.7 &71.4& 2.6 \\
  & knn+cliff & 60.0 &83.3 &2.6 \\
  & knn+cnn & 50.0 &52.9 &4.0 \\
  & knn+mcs & 50.0 &50.0 &5.0 \\ 
  & knn+psc & 35.7 &33.0& 8.8  \\ \hline
\multirow{5}{HeartC} & knn & 76.7 &75.0& 23.1  \\
  & knn+cliff & 82.6& 83.3& 17.2 \\
  & knn+cnn & 74.2 &75.0 &25.7 \\
  & knn+mcs & 54.1 &53.3& 43.8 \\ 
  & knn+psc & 50.0 &50.0& 50.0  \\ \hline
\multirow{5}{Hepatitis} & knn & 83.3& 82.6 &16.7  \\
  & knn+cliff & 81.8& 77.8 &17.4 \\
  & knn+cnn & 77.8 &71.4& 21.7 \\
  & knn+mcs & 63.6 &68.8& 34.6 \\ 
  & knn+psc & 60.9 &66.7& 38.5  \\ \hline
\multirow{5}{Iris} & knn & 100.0 & 100.0 & 0.0  \\
  & knn+cliff & 100.0 & 100.0 & 0.0 \\
  & knn+cnn & 90.9& 88.9& 4.3 \\
  & knn+mcs & 90.9 &90.0& 4.5 \\ 
  & knn+psc & 90.9 &92.3 &4.3 \\ \hline
\multirow{5}{Labor} & knn & 100.0 & 100.0 & 0.0  \\
  & knn+cliff & 88.9 &88.9& 11.1 \\
  & knn+cnn & 90.9 &85.7& 0.0 \\
  & knn+mcs & 100.0 & 100.0 & 0.0 \\ 
  & knn+psc & 100.0 & 100.0 & 0.0  \\ \hline
\multirow{5}{Lymph} & knn & 78.6& 78.9& 0.0  \\
  & knn+cliff & 76.9& 70.0& 6.7 \\
  & knn+cnn & 66.7 &66.7& 0.0 \\
  & knn+mcs & 68.8 &70.6& 0.0 \\ 
  & knn+psc & 58.8 &63.6& 3.4  \\ \hline
\multirow{5}{Vote} & knn & 92.1& 92.2& 7.7  \\
  & knn+cliff & 89.8& 88.1& 9.8 \\
  & knn+cnn & 88.9 &88.2 &11.1 \\
  & knn+mcs & 70.4 &72.9& 29.3 \\ 
  & knn+psc & 72.2 &71.4& 27.0  \\ \hline
\end{tabular}}
\caption{pd, prec and pf results for Experiment \#1}
\label{fig:res1}
\end{center}
\end{figure}
\end{comment}

%\subsection{Is CLIFF more tolerant to noise than other IS?}



%With that said, in this experiment we repeat Experiment \#1, only this time noise is introduced to training data by randomly changing the target class of 10\% of the instances in the training data to any other target class value. The the results here will indicate how well the 1NN classifier along with the different IS are able to predict the correct target class even if some of its training data is faulty \cite{wilson00}. 

%\subsection{Results from Experiment \#2}
%The $noisy$ tables in \fig{noise} displays the results of Experiment \#2. The first thing to recognize is that compared to the $clean$ tables, there is a general degradation of the pd and pf results for each data set. However in some cases such as the pd results for $bc$ the pds can degrade by as little as 1\%. The second thing to recognize is that while the $sizes$ of the training data basically remains the same for CLIFF (differences of no more that 2\%) for all data sets, the other IS display a general increase in the training set sizes except for PSC whose training size decrease from 42\% to 40\%. 
\begin{comment}
\begin{figure*}[!ht]
\begin{center}
\renewcommand{\baselinestretch}{0.7}
\begin{tabular}{p{0.25in}@{}c@{}c}
&KNN & CLIFF \\ \hline
pd&\includegraphics[width=2.5in]{knnpd.pdf} &
\includegraphics[width=2.5in]{cliffpd.pdf}
\\\hline
prec&\includegraphics[width=2.5in]{knnprec.pdf} &
\includegraphics[width=2.5in]{cliffprec.pdf}
\\\hline
pf&\includegraphics[width=2.5in]{knnpf.pdf} &
\includegraphics[width=2.5in]{cliffpf.pdf}
\\
\end{tabular}
\end{center}
\caption{Effects of noise.}
\label{fig:noise}
\end{figure*}
\end{comment}
%%%%%%%%%%%%%

\begin{figure*}[!ht]
\begin{center}
\renewcommand{\baselinestretch}{0.5}
\begin{tabular}{p{0.20in}@{}c@{}c@{}c}
&pd & prec & pf \\ \hline
KNN&\includegraphics[width=2in]{knnpd.pdf} &
\includegraphics[width=2in]{knnprec.pdf} &
\includegraphics[width=2in]{knnpf.pdf} &
\\\hline
CLIFF&\includegraphics[width=2in]{cliffpd.pdf} &
\includegraphics[width=2in]{cliffprec.pdf} &
\includegraphics[width=2in]{cliffpf.pdf}
\\
\end{tabular}
\end{center}
\caption{Effects of noise: Performance measures (pd, prec and pf) of CLIFF vs. KNN in the presence of noise.}
\label{fig:noise}
\end{figure*}

%%%%%%%%%%%%%

\begin{figure}[!ht]
\begin{center}
\renewcommand{\baselinestretch}{0.6}
{\scriptsize
\begin{tabular}{lr|rrrrr}
& & \multicolumn{5}{c}{percent noise}\\
& & 0 & 10 & 20 & 30 & 40\\\hline
\input{pds}
\end{tabular}
}
\caption{PDS: other - (knn+cliff). Note: {\em negative} values means CLIFF is performing {\em better}.} 
\label{fig:better1}
\end{center}
\end{figure}

\begin{figure}[!ht]
\begin{center}
\renewcommand{\baselinestretch}{0.6}
{\scriptsize
\begin{tabular}{lr|rrrrr}
& & \multicolumn{5}{c}{percent noise}\\
& & 0 & 10 & 20 & 30 & 40\\\hline
\input{precs}
\end{tabular}
}
\caption{PRECISION: other - (knn+cliff). Note: {\em negative} values means CLIFF is performing {\em better}.} 
\label{fig:better2}
\end{center}
\end{figure}

\begin{figure}[!ht]
\begin{center}
\renewcommand{\baselinestretch}{0.6}
{\scriptsize
\begin{tabular}{lr|rrrrr}
& & \multicolumn{5}{c}{percent noise}\\
& & 0 & 10 & 20 & 30 & 40\\\hline
\input{pfs}
\end{tabular}
}
\caption{PFs: other - (knn+cliff). Note: {\em positive} values means CLIFF is performing {\em better}.} 
\label{fig:better3}
\end{center}
\end{figure}
\begin{comment}
\subsection{Is CLIFF better at reducing brittleness than other IS?}

In this work, $brittleness$ refers to the following:

\begin{quote}
\emph{Brittleness} is a measure of whether a solution (predicted target class) comes from a region of similar solutions or from a region of dissimilar solutions. Or, looking at this another way, how far would a test instance have to move before a different target class is predicted.
\end{quote}


\begin{figure}[ht!]
  \begin{center}
  \scalebox{1}{
    \begin{tabular}{l}
      \resizebox{100mm}{!}{\includegraphics{ir2fss}} \\
      \resizebox{100mm}{!}{\includegraphics{ir2fssaf}}      
    \end{tabular}}
    \caption{}
    \label{fig:charts2}
  \end{center}
\end{figure} 

Take for example the \emph{Before CLIFF} chart in \fig{charts2}, the classes $versicolor$ and $virginica$ obviously show severe overlap. Also, the versicolor test instance represented by the purple square does not have to move very far before it can change its predicted target class to $virginica$. Looking now at the \emph{After CLIFF} chart in \fig{charts2}, after applying CLIFF, a subset of instances are selected as prototypes. This increases the distance from the $versicolor$ test instance to a prototype with the $virginica$ class thereby reducing $brittleness$.

With this measure in mind, the goal here is to see how each prototype learning scheme studied here reduces the brittleness of the KNN model where k=1. Brittleness will be measured by distance each test instance moves before changing its original predicted target class. Intuitively, the further away the test instance has to move the less $brittle$ the model. The experiment design for $brittleness$ can be done in conjunction with Experiment 1 by collecting the distances of each test instance with a predicted target class to the nearest training instance with a different target class. This is done for each prototype learning scheme and 1NN. The distances generated are joined, sorted and labelled according to their position in the list. for example, let us say that A and B are IS with the distance values of [2, 2, 2, 3, 4, 3, 77] and [6, 7, 3, 9, 1, 1, 1, 100] respectively. After being joined, sorted and labelled the result is as follows:

\begin{figure}[ht!]
\scalebox{0.7}{
\begin{tabular}{l*{14}{c}r}
IS  & B & B & B & A  & A & A & A & A & B &  A &  B &  B &  B &  A  &  B \\
\hline
Sort & 1 & 1 & 1 & 2 & 2 & 2 & 3 & 3 & 3 &  4 &  6 &  7 &  9 & 77 & 100  \\
Position & 1 & 2 & 3 & 4 & 5 & 6 & 7 & 8 & 9 & 10 & 11 & 12 & 13 & 14 &  15  \\
Label & 1 & 1 & 1 & 5 & 5 & 5 & 8 & 8 & 8 & 10 & 11 & 12 & 13 & 14  & 15  \\
\end{tabular}}
\end{figure}
	
As shown, labels are assigned according to the position of a value in the list, however, if values are the same, the mean of their position values are used as a label.

\subsection{Results from Experiment \#3}
\fig{charts1} and \fig{instances} present the results for this experiment. The pattern is very clear: CLIFF does a much better job of reducing $brittleness$ in all cases than any of the other IS. Each chart in \fig{charts1} represents results for the different data sets used in this work. They all show that the CLIFF test instances have to move further away (most of the time) before there is a change in their target classes. These results are confirmed by a Mann Whitney statistical test (\fig{instances}), which show that the CLIFF results are statistically different and better than the other IS. 
\end{comment}


\subsection{Summary}
Collectively, the results of the above experiment indicates that CLIFF may be effective in the field of forensic interpretation where a very low false alarm rate $pf$ is desired. Here, CLIFF can be used to help lower the $pf$s of a forensic interpretation model. The results shown in \fig{noise} suggest this possibility in the median $pf$ results where CLIFF's values ranges from 0 to 47\% and are lower or the same as the baseline (KNN) $pf$ results.

\begin{comment}
\begin{figure*}[ht!]
%  \begin{center}
  \scalebox{0.87}{
    \begin{tabular}{l}
      \resizebox{90mm}{!}{\includegraphics{bc1}}
      \resizebox{90mm}{!}{\includegraphics{dm1}} \\
      \resizebox{90mm}{!}{\includegraphics{hc1}} 
      \resizebox{90mm}{!}{\includegraphics{hh1}} \\
      \resizebox{90mm}{!}{\includegraphics{ir1}} 
      \resizebox{90mm}{!}{\includegraphics{lv1}} \\
      \resizebox{90mm}{!}{\includegraphics{mm1}} \\
    \end{tabular}}
    \caption{Position of distance values for IS}
    \label{fig:charts1}
 % \end{center}
\end{figure*}
\end{comment}
